Author's response to reviews

Title: Association of intradialytic hypotension and convective volume in hemodiafiltration: results from a retrospective cohort study.

Authors:

Franklin G Mora-Bravo (franklin.mora.bravo@gmail.com)
Guadalupe De-La-Cruz (gpe.delacruz.2@gmail.com)
Sonia Rivera (cati.rivera.gonzalez@hotmail.com)
Alfonso Mariscal-Ramírez (mariscalmd@yahoo.com.mx)
Jochen G Raimann (JRaimann@rriny.com)
Héctor Pérez-Grovas (hpgrovas@gmail.com)

Version: 2 Date: 11 February 2012

Author's response to reviews:

Dear
Wella Valenzuela
Journal Editorial
Office Biomed Central

Through this letter I inform you that the following changes were made:
1. We add the bioethics committee approval
2. According to the guidelines format tables were placed after the references
3. was the format the entire manuscript according to the guidelines of the Journal.

Sincerely
Franklin Mora-Bravo